vc/22-Sep-03
Week 39 starting 22 Sept 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(23/09/03)
	Forward look,   & Next Magnet 

	TBA1
	2012
	Should be Warming Up
	Bench to be de-activated – preparation for D bench

	TBA2
	3042
	Under Connection
	 Cold Tests Start Thurs night?

	TBC1
	2021
	Need to connect
	Cooldown by Fri.Start Cold Tests start on Fri Night during w’end

	TBC2
	SSS1
	cold
	Cold tests

	TBE1
	3043
	Warm 
	Removal – bring new 2014

	TBE2
	2010 
	cold
	Cold tests  

	TBF1
	2025 after F2 done? 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025 proposed
	
	Prepare For 

Commissioning FIRST


Test Coordinator this week:  N Smirnov;  MTM  Technician on call:  (16-3602) : 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. New additional & easy  Tel nos.   70900  as well as 16 0060

2. Paper Logbook  & Files 

· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 16_M


	TBE 1 : TEMA cannot be launched
	AB/CO
	TEMA
	Script, which was managing the conflicts between applications, gave errors. Solved by Pedro.

	Tuesday 16_A
	TBE1 : IAP@cold 

Errors in compensation out of the limits defined for one channel.
	MTM
	
	Accepted by Giorgio.

Origin not found.



	Tuesday 16_A
	TBE1: IDCCT current reading not working

Problem in LHC cycle

Load-line without read current.


	AB/CO
	
	Solved the 17 morning

DDCT replaced by DDCT of cluster B .

No calibration so read-out connected in the bottom of the DDCT.

	Tuesday 16_N
	TBE1 : IAP@warm cannot be performed


	MTM
	
	Connector inverted

	Tuesday 16_N
	MM : TBA2 MUX errors


	AB/CO
	
	

	Tuesday 16_N
	TBA1 : IAP @ warm could not be performed.
	MTM
	
	Cards missing in HF/LF crates used by cluster F.

	Wednesday 17_M
	Full load line (TBE1): wrong current reading 
	
	
	 Run recovery and then OK

Origin not found

	Thursday 18_M
	TBE1: cable coupling current MM5 
	MTM
	TEMA?
	Wrong cycle selected

Magnetiz

	Saturday 20_M

Sunday 21_M
	TBA1: unable to launch IAP cold_X2 from TEMA
	AB/CO
	IAP
	IAP done manually.

Problem under investigation

	Saturday 20_A
	TBE2: Tanticrysotat no visible in the control room 

Communication problem between PLCART and cryo supervision 
	MTM/ACR
	
	Local check of the good functioning of the anticryostat.

	Saturday 20_N
	TBE1: Rotary switch blocked
	MTM
	
	The handling of the switch was not correctly done.

See procedure.

	Saturday 20_N
	TBE1 : IAP at cold. One card faulty  (8 channels out of range) in aperture 2
	MTM
	IAP
	

	Saturday 20_N

Sunday 21_M
	TBA1 : Vi corrupted in the MMP version because of an update of the library which was not working in TBA1


	AB/CO
	MMP
	Adriaan called : Launched the old version of MMP.

Library updated Monday morning

	Sunday 21_M
	New PC : No macro available for the quench table.
	MTM
	
	Security setting corrected

Solved (Daniel)



	Sunday 21_M
	Vclamp trigger could not be cleared in TBC1????
	MTM
	
	????



	Sunday 21_N
	MM in TBA1 launched with MMP old version

Full load line: Motor in aperture 1 is turning but not the shaft.
	AB/CO
	
	Then communication error with one motor during auto test.

Solved by switching to another TRU and back on display???



	Monday 22_M
	Full load line in TBA1: problem with end stops
	MTM
	
	

	Monday 22_M
	Still problems with anti cryostat reading in the control room
	MTM/ACR
	
	Not solved.

Local check



	Monday 22_M
	FTP errors: unable to launch TEMA
	
	
	Automatic update of the FTP at CERN 

	Monday 22_M
	TBE2: MM positioning

All the sectors with b11 in orange and MM position OK?
	MTM
	
	AP1/AP2 connectors mixed when connected to TRU 

Table for MM errors in display .

	Monday 22_A
	TBE2: LHC cycle

End field polarity error not seen
	MTM
	
	Switch on a instead of e coil in 0001 for shaft 25. 


